Author’s response to reviews

Title: Lavender oil suppresses indoleamine 2,3-dioxygenase activity in human PBMC

Authors:

- Johanna M Gostner (johanna.gostner@i-med.ac.at)
- Markus Ganzera (Markus.Ganzera@uibk.ac.at)
- Kathrin Becker (kathrin.becker@i-med.ac.at)
- Simon Geisler (simon.geisler@i-med.ac.at)
- Sebastian Schröcksnadel (sebastian.schroecksnadel@i-med.ac.at)
- Florian Überall (florian.ueberall@i-med.ac.at)
- Harald Schennach (harald.schennach@uki.at)
- Dietmar Fuchs (dietmar.fuchs@i-med.ac.at)

Version: 2
Date: 13 January 2014

Author’s response to reviews:

Dear Dr. Prozenko,

following changes were made to meet the editorial requirements:

1) The study was performed according the Helsinki Declaration. This is stated now in the method section.
2) The contribution of each author was added.

We hope that the manuscript is now acceptable for peer review.

Yours sincerely,

Johanna Gostner